RESEARCH ARTICLE

# Laparoscopic resection of hepatic alveolar echinococcosis: A single-center experience

#### 

Department of Visceral Surgery and Medicine, Inselspital, Bern University Hospital, University of Bern, Switzerland

\* anja.lachenmayer@insel.ch

# Abstract

# Introduction

Alveolar echinococcosis (AE) remains a very rare disease requiring complete radical resection for curative treatment. While open approaches are common, safety and efficacy of laparoscopic resections remain unknown.

# Methods

This is a single-center, retrospective cohort study with patients undergoing liver resection for hepatic AE at the Department of Visceral Surgery and Medicine, Bern University Hospital from December 2002 to December 2020. Postoperative outcomes of patients following laparoscopic hepatectomy (LH) for hepatic AE were compared with those of patients undergoing open hepatectomy (OH).

# Results

A total of 93 patients underwent liver resection for hepatic AE. Laparoscopic hepatectomy was performed in 23 patients and open hepatectomy in 70 patients. While there were no significant differences in terms of gender, age and diagnostic tools, the majority of patients of the LH cohort were PNM stage 1 (78%) in contrast to only 39% in the OH cohort (p = 0.002). Patients undergoing laparoscopic hepatectomy were treated by minor liver resections in 91% and in 9% by major liver resections in comparison to the open hepatectomy was associated with shorter mean operation time (127 minutes vs. 242 minutes, p < 0.001), lower major complication rate (0% vs. 11%, p = 0.322) and shorter mean length of hospital stay (4 days vs. 13 days, p < 0.001). Patients with LH had a distinct, but not significant lower recurrence rate (0% vs. 4%, p = 0.210) during a mean follow-up of 55 months compared with a follow-up of 76 months in the OH cohort. After subgroup analysis of PNM stage 1 patients, similar results are seen with persistent shorter mean operation time (120 minutes vs. 223 minutes, p < 0.001), lower major complication rate (0% vs. 12 days, p < 0.001).

# OPEN ACCESS

**Citation:** Gloor S, Candinas D, Beldi G, Lachenmayer A (2022) Laparoscopic resection of hepatic alveolar echinococcosis: A single-center experience. PLoS Negl Trop Dis 16(9): e0010708. https://doi.org/10.1371/journal.pntd.0010708

Editor: Thomas F. E. Barth, University of Ulm, GERMANY

Received: February 3, 2022

Accepted: July 29, 2022

Published: September 6, 2022

**Copyright:** © 2022 Gloor et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the manuscript.

**Funding:** The authors received no specific funding for this work.

**Competing interests:** The authors have declared that no competing interests exist.

#### Conclusion

Laparoscopy appears as a feasible and safe approach for patients with PNM stage 1 alveolar echinococcosis without impact on early disease recurrence.

#### Author summary

In this retrospective cohort study laparoscopic hepatectomy for hepatic alveolar echinococcosis had no negative impact on perioperative outcomes, disease recurrence or survival compared with open hepatectomy. The importance of this finding is that the laparoscopic approach is feasible and safe for selected patients with hepatic alveolar echinococcosis, especially those with PNM stage 1.

# Introduction

Echinococcus multilocularis is the causative agent of alveolar echinococcosis (AE) which predominantly affects the liver [1]. Only in 2.3% of cases other organs (e.g. diaphragm, peri-renal tissue, lymph nodes, peritoneum, lung, brain, spleen or bone) become initially infected by the larval stages (metacestodes) of the parasite [2,3]. Although its overall incidence remains low, certain endemic regions report increasing numbers including Switzerland with an incidence of 0.3 to 3/million inhabitants/year [2,4].

About 70% of patients typically present with symptoms such as hepatobiliary complications or abdominal pain, while one third of patients is usually detected incidentally [5,6]. With the increasing use of cross sectional imaging (abdominal sonography, computed tomography or magnetic resonance imaging), immunosuppressive medication and eventually also an altered infectious potential, diagnoses seem to increase in the recent years [4]. Further, serology (purified Echinococcus multilocularis Em2-ELISA, Echinococcus multilocularis recEm18-ELISA and Echinococcus granulosus EgHF-ELISA) of AE supports differentiation of hepatic lesions, enables to differentiate between alveolar and cystic echinococcosis [7] and to interpret the activity of the infection [8].

The only curative treatment available is a complete resection of the part of the liver containing the parasitic mass. Radical resection with open surgery is up to now the standard approach for hepatic AE-lesions <sup>5</sup> and several studies have shown safety and efficacy with rates of moderate and major complications (Clavien-Dindo  $\geq$ 3a) between 2.7% and 15% [9–11]. While cystic echinococcosis is now increasingly operated laparoscopically [12], very few studies have reported feasibility and safety of laparoscopic and robotic resection in AE patients so far [13– 15]. Although laparoscopic hepatic resections are now commonly used for hepatic malignancies with no differences in oncologic results tested on colorectal liver metastasis [16], not much is known about the safety and efficacy of laparoscopic surgery for AE. Especially still recommended safety margin of 2 cm in order to avoid disease recurrence because of its slow and progressive infiltrative growth into liver tissue and along the central bile ducts [17] inhibited progress of laparoscopic AE resection.

Current recommendations suggest to treat with Albendazole (ABZ) 2 years after complete surgical resection [5]. In patients with non-resectable disease lifelong ABZ-therapy offers excellent long-term survival [18,19] with a low incidence of severe adverse effects of ABZ-therapy such as alopecia, elevated liver enzymes, gastrointestinal disorders or leucopenia [1,20,21].

We therefore aimed to analyze safety and feasibility of laparoscopic resection for alveolar echinococcosis in our retrospective single-center analysis.

# Methods

#### **Ethics statement**

This study was initiated after obtaining approval from the Ethics Commission of the Canton of Bern (KEK Nr. 2017–01534) and with an amendment for the time frame 2018–2020. General written informed consent was obtained from all subjects involved in the study.

#### Patient inclusion criteria

This trial is a single-center, retrospective analysis of patients undergoing liver resections due to alveolar echinococcosis (AE) at the Department of Visceral Surgery and Medicine at the Inselspital, Bern University Hospital.

In our clinic a total of 155 patients were treated due to AE during the period of 2002–2020. The inclusion criteria contained adult patients over age 18, which underwent elective liver resection for AE. Patients not requiring biliary reconstructions or large anatomic resections such as hemihepatectomies, extended hemihepatectomies or central liver resections were oper-ated laparoscopically if technically feasible. The cohort of 62 patients who were only treated with ABZ were excluded. None of the patients had documented objection to subsequent use of personal health data. The data of the conservatively treated patients has recently been published [22]. Clinicopathological data of 93 consecutive patients were extracted including demographics, radiological characteristics, therapeutic features, intraoperative findings, surgical procedures, pathological examination, morbidity and complications.

#### **Preoperative assessment**

All patients suffering from AE were evaluated by an interdisciplinary team of hepatologists and hepatobiliary surgeons according to the standardized procedure (medical history, physical examination, laboratory tests, radiological exams and anesthesia evaluation) in our department. PNM stage was defined from radiological exams [23], either by computed tomography or magnetic resonance imaging. Serological testing of Echinococcus-specific antibodies (EgHF, Em2 and recEm18) was used as an additional examination to diagnose AE. Individual treatment for each patient was discussed in the weekly multidisciplinary liver boards attended by hepatobiliary surgeons, hepatologists, radiologists and pathologists. Hepatectomy was recommended if AE lesions could be resected with preservation of a sufficient future liver remnant (FLR). To assess lesion location in the liver and to estimate FLR contrast-enhanced computed tomography (CT) or magnetic resonance imaging (MRI) with liver-specific contrast agents was performed. If FLR was estimated smaller than 30% of liver volume, 3D-volumetry was added to calculate exact volumes and to define risk of liver insufficiency. Preoperative medical therapy with albendazole (ABZ) was administered according to the assessment of the multidisciplinary liver boards. Patients were excluded, if no curative surgical resection could be performed or patients refused the operation.

# Surgical procedure and postoperative management

Patients were treated by laparoscopy according to the preoperative decision of the surgeon which was based on the experience with laparoscopic liver surgery of primary or secondary liver malignancies [16]. The difficulty score for laparoscopic liver resection was calculated [24]. This score includes tumor location, extent of liver resection, tumor size, proximity to

major vessel and the liver function and reflects difficulty of laparoscopic approach well. All liver resections were performed by AE-experienced and hepatobiliary surgeons also trained and experienced in oncologic laparoscopic liver surgery. Blood loss was calculated after suction of all intraabdominal liquid during operation, whereas flushing liquid was subtracted afterwards. Postoperative complications within 90 days after surgery were graded according to the classification of Clavien and Dindo [25]. Major morbidity was defined as any complication  $\geq$ 3b grade and postoperative mortality as grade 5.

Patients received disease surveillance according to institutional standards and international guidelines at least annually [5]. Postoperative medical therapy with ABZ was administered according to the postoperative assessment of the multidisciplinary liver boards. AE recurrence was defined as the first lesion relapse following curative intended hepatectomy for AE combined with an increase of sampled antibodies in serology. Radiologists reviewed during the scheduled long-term disease surveillance cross-sectional imaging (CT-scan, MRI) to identify the presence and location of recurrent disease.

# Histological evaluation

Histology was performed upon resected specimens, yielding diagnosis and resection margin status of AE, as well as resection margin. R1 resection was defined as the microscopic presence and R2 as the macroscopic presence of the laminar layer within the transection line, whereas R0 resection was defined as complete resection with no laminar layer within 1 mm of the resection margin. In case of R2 resection any detected disease of AE during follow-up was classified as persistent.

# Serological testing

From all patients serum samples were collected to test perioperative Echinococcus-specific antibodies (EgHF, Em2 and recEm18). All three ELISAs were carried out as previously described [26]. Antibody levels against EgHF, Em2 and recEm18 were compared between last preoperative and first postoperative samples, which could vary due to missing international standards.

# Statistical analysis

Quantitative and qualitative variables were expressed as mean (standard deviation) and frequency (percentage). Number and stage of complications as well as overall survival (OS), recurrence-free survival (RFS), recurrence rate, length of medical therapy and mortality were analyzed. Comparisons between cohorts were analyzed with the chi-square or Fisher exact test for categorical variables and the Mann-Whitney U test for continuous variables, as appropriate. Using the Kaplan-Meier method, OS and RFS was calculated from the date of resection to the date of death, date of recurrence or last follow-up respectively. Comparisons between survival or recurrence rates were performed using log-rank tests. Survival time was calculated from the date of operation until the date of interest (last visit or death). P < 0.05 was considered statistically significant. Subgroup analyses were performed for patients with PNM stage 1. The subgroup is labelled with the prefix 'PNM stage 1'. Statistical analysis was performed using SPSS version 25 (IBM, Armonk, New York, USA).

#### Results

#### **Clinical data**

Table 1 summarizes the clinical data of the 23/93 (25%) patients with laparoscopic hepatectomy (LH) and 70/93 (75%) with open hepatectomy (OH). While there were no significant

Variable	LH (n = 23)	OH (n = 70)	p-value
Gender, n (%)			0.852
Female	13/23 (57)	38/70 (54)	
Male	10/23 (43)	32/70 (56)	
Age, mean years (SD)	57 (±17)	53 (±15)	0.193
Primary radiologic examination, n (%)			
Sonography	3/23 (13)	9/70 (13)	0.982
CT-scan	14/23 (61)	43/70 (61)	0.962
MRI	4/23 (17)	8/70 (11)	0.462
Unknown	2/23 (9)	10/70 (15)	0.490
PNM-stage, n (%)			
1	18/23 (78)	25/70 (39)	<0.001
2	0/23 (0)	11/70 (16)	0.044
3a	3/23 (13)	10/70 (14)	0.882
3b	2/23 (9)	14/70 (20)	0.215
4	0/23 (0)	7/70 (10)	0.117
Unknown	0/23 (0)	3/70 (4)	0.315
Maximal diameter of lesion, mean cm (SD)	4.3 (±3.1)	8.6 (±4.7)	<0.001
Local extrahepatic lesions, n (%)	1/23 (4)	14/70 (20)	0.042
Distant extrahepatic lesions, n (%)	0/23 (0)	2/70 (3)	0.408
Preoperative treatment with albendazole, n (%)	16/23 (70)	51/70 (73)	0.392
Time from diagnosis to operation, mean months (SD)	4 (±5)	5 (±13)	0.554

Table 1. Comparison of demographic and preoperative clinical data between laparoscopic and open resection of patients with hepatic alveolar echinococcosis.

CT, computed tomography; LH, Laparoscopic hepatectomy; MRI, magnetic resonance imaging; OH, open hepatectomy; PNM, PNM system (P = parasitic mass in the liver, N = involvement of neighbouring organs, and M = metastasis); SD, standard deviation

https://doi.org/10.1371/journal.pntd.0010708.t001

differences in terms of gender, age and diagnostic tools, the majority patients of the LH cohort were PNM stage 1 (n = 18/23, 78%) in contrast to only 39% (n = 25/70) in the OH cohort (p = 0.002). The hepatic lesions in the LH cohort were located in one left-sided segment (n = 4/23, 17%), one central segment (n = 1/23, 4%), one right sided segment (n = 2/23, 9%), two left-sided segments (n = 2/23, 9%), two central segments (n = 2/23, 9%), two right-sided segments (n = 2/23, 9%), more than three right-sided segments (n = 3/23, 13%) and more than three bilateral segments (n = 7/23, 30%). In contrast, hepatic lesions of the OH cohort were located in one left-sided segment (n = 1/70, 1%), one central segment (n = 3/70, 4%), one right sided segment (n = 6/70, 9%), two left-sided segments (n = 5/70, 7%), two central segments (n = 2/70, 3%), two right-sided segments (n = 12/70, 17%), two bilateral segments (n = 2/70, 17%)3%), more than three left-sided segments (n = 7/70, 10%), more than three central segments (n = 4/70, 6%), more than three right-sided segments (n = 13/70, 19%) and more than three bilateral segments (n = 15/70, 21%). The mean maximal diameter was significantly higher (8.6 cm vs. 4.3 cm, p < 0.001 in the OH cohort. Local extrahepatic lesions were more common in OH cohort (OH 20% [n = 14/70] vs. LH 4% [n = 1/23], p = 0.042) and consisted of infiltration to the diaphragm and the greater omentum. A distant extrahepatic disease to brain (n = 1) and peritoneum (n = 1) existed in two patients of the OH cohort (n = 2/70, 3%), compared to absent extrahepatic disease in the LH cohort (p = 0.408). The interdisciplinary tumor board has recommended a resection of that brain metastasis due to neurological symptoms after primary hepatectomy. The other patient received a tumor debulking surgery in palliative intention because he suffered from severe edemas of the lower limbs and massive ascites due to an

external compression of the inferior vena cava and the portal vein caused by massive peritoneal AE, respectively. Further there was an insignificant difference for the mean time from diagnosis to resection, which was shorter in the OH cohort (4 months vs. 5 months, p = 0.554) (Table 1).

#### Perioperative data

The majority of patients in the LH cohort were treated by minor hepatic resections (n = 21/23, 91%) including bi-segmentectomies (n = 4/23, 17%), segmentectomies (n = 1/23, 4%) or atypical liver resections (n = 16/23, 70%) in comparison to the OH cohort with significant more major liver resections and only 39% (n = 27/70) minor resections (13% bi-segmentectomies [n = 9/70], 6% segmentectomies [n = 4/70] or 20% atypical liver resections [n = 14/70]). After the first laparo-scopic AE resection at our center in 2006 the caseload increased steadily within the years (Fig 1). According to the difficulty score for laparoscopic liver resection a mean score of 5.05 (SD ±2.8) points in the LH cohort was seen [24]. The mean operation time was significantly shorter in the LH patients compared with OH patients (127 minutes vs. 242 minutes, p < 0.001) and LH patients have lost less blood during operation (307 ml vs. 694 ml, p = 0.003). Conversion from laparoscopic to open approach was needed in two patients due to technical issues in case of tumor extent across the central bile duct in one case and the first portal pedicle in the other case requiring open dissection and biliary reconstruction. Fig 2 shows two examples of laparoscopic AE resections.

# Complications and hospital stay

The mean length of stay in the LH cohort was significantly shorter (4 days) than in the OH cohort (13 days, p < 0.001) (Table 2). The LH cohort had a significantly lower overall



ig 1. Development of caseload in AE conort.

https://doi.org/10.1371/journal.pntd.0010708.g001



**Fig 2.** Laparoscopic resection of AE-lesions in two patients: In the first patient, CT-scan shows a small lesion in central liver segment 4a/b (A). Prior to resection an intraoperative sonography was made and the lesion was marked (B). The dissection of liver parenchyma was performed by an ultrasonic dissection device (C). A post-resectional sonography showed free resection margins (D). The CT-scan of a second patient shows a calcified lesion in liver segment 4b (E). This lesion could be macroscopically seen in the diagnostic laparoscopy (F). Dissection of liver parenchyma was done by an ultrasonic dissection device (G) and by ultrasonic surgical aspirator system (H).

complication rate (9% [n = 2/23] vs. 43% [n = 30/70], p = 0.003) and lower major complication rate (0% [n = 0/23] vs. 11% [n = 8/70], p = 0.322). In the LH cohort the only minor complication of grade 3a according to Clavien-Dindo requiring intervention was a biliary leakage from the resection area treated with percutaneous drain insertion and stenting of the biliary duct by endoscopic retrograde cholangiopancreaticography. In the OH cohort, 8 major complications occurred including five infected biliomas, an acute portal vein thrombosis, an acute bleeding requiring operative therapy and an acute renal insufficiency requiring hemodialysis and intensive care.

#### Serology

Serum antibody levels (EgHF [n = 72/93, 77%], Em2 [n = 78/93, 84%] and recEm18 [n = 76/ 93, 82%]) were significantly lower within 12 months after curative resection. Due to missing data, in particular in the earlier cases where serology was not routinely used, not all patients could be examined. recEm18, which has the most clinical use of the mentioned antibodies, presented a negativization rate of 65.7% (n = 50/76) during postoperative course. This was apparent in LH cohort as well as in OH cohort and is displayed in Fig 3. LH cohort had lower mean preoperative values in antibodies than OH cohort (EgHF: 57 AU/ml vs. 77 AU/ml, p = 0.064; Em2: 18 AU/ml vs. 40 AU/ml, p = 0.017; recEm18: 19 AU/ml vs. 38 AU/ml, p = 0.018).

#### Perioperative treatment with albendazole

Albendazole (ABZ) was administered as medical therapy in patients receiving pre- or postoperative benzimidazole therapy. In both cohorts a similar percentage of patients received preoperative medication with ABZ (LH 70% [n = 16/23] vs. OH 73% [n = 51/70], p = 0.392) (Table 1). Postoperatively, patients with LH received less often postoperative therapy (n = 17/ 23, 74%) compared with OH patients (n = 64/70, 93%, p = 0.044), as it is shown in Table 2. The mean duration of application of preoperative ABZ-medication was shorter in the LH cohort (3 months [SD ±2]) compared with patients from the OH cohort (5 months [SD ±9], p = 0.529). Postoperative medical therapy in mean was not significantly shorter in patients treated laparoscopically (LH 15 months [SD ±10] vs. OH 22 months [SD ±25], p = 0.684). In total, 10 patients (n = 10/93, 11%) developed adverse reactions to the medical therapy with

Variable	LH (n = 23)	OH (n = 70)	p-value
Type of resection, n (%)			
Hemihepatectomy right	2/23 (9)	17/70 (24)	0.110
Extended hemihepatectomy right	0/23 (0)	14/70 (20)	0.021
Hemihepatectomy left	0/23 (0)	8/70 (11)	0.092
Extended Hemihepatectomy left	0/23 (0)	2/70 (3)	0.415
Bisegmentectomy	4/23 (17)	9/70 (13)	0.588
Segmentectomy	1/23 (4)	4/70 (6)	0.802
Atypical resection	16/23 (70)	14/70 (20)	<0.001
Other	0/23 (0)	2/70 (3)	0.415
Operation time, mean minutes (SD)	127 (±60)	242 (±98)	<0.001
Blood loss, mean ml (SD)	307 (±327)	694 (±516)	0.003
Complications (Clavien-Dindo), n (%)	2/23 (9)	30/70 (43)	
1	0/23 (0)	1/70 (1)	0.567
2	1/23 (4)	8/70 (11)	0.322
3a	1/23 (4)	15/70 (21)	0.016
3b	0/23 (0)	4/70 (6)	0.802
4a	0/23 (0)	4/70 (6)	0.244
4b	0/23 (0)	0/70 (0)	-
5	0/23 (0)	0/70 (0)	-
Major complication (Clavien-Dindo $\geq$ 3b), n (%)	0/23 (0)	8/70 (11)	0.322
Length of hospital stay, mean days (SD)	4 (±2)	13 (±10)	<0.001
Postoperative treatment with albendazole, n (%)	17/23 (74)	64/70 (93)	0.044
Resection status, n (%)			
RO	16/23 (70)	39/70 (56)	0.244
R1	5/23 (22)	19/70 (27)	0.609
R2	0/23 (0)	3/70 (4)	0.315
Unknown	2/23 (9)	9/70 (13)	0.594
Recurrence, n (%)	0/23 (0)	3/70 (4)	0.210

LH, Laparoscopic hepatectomy; OH, open hepatectomy; SD, standard deviation

https://doi.org/10.1371/journal.pntd.0010708.t002





ABZ. Increased liver transaminases were seen in 4 patients and 6 patients suffered from alopecia.

#### Histopathological data

In the histopathological samples analyzed (n = 82/93, 88%), AE was confirmed in all cases, although PAS activity was only reported in 33/82 patients (40%). No significant differences were detected for the rate of R1-resections (LH 22% [n = 5/23] vs. OH 27% [n = 19/70], p = 0.265). The mean resection margin in R0 resected patients was also not significantly different (LH 0.3 cm [±0.4 cm, 13/16] vs. OH 0.7 cm [±1.1 cm, 34/39], p = 0.605) However, resection margin ranged from 0.0 cm to 5.0 cm in the OH cohort and from 0.0 cm to 2.0 cm in the LH cohort, and exact numbers were missing for 3 and 5 samples, respectively.

#### Follow-up and recurrence

The 5-year overall survival showed no differences (LH 100% vs. OH 97%, p = 0.264), as it is seen in Fig 4A. Patients with LH had a distinct, but not significant lower recurrence rate (0% [n = 0/23] vs. 6% [n = 3/70], p = 0.210) during a mean follow-up of 55 months compared with



Fig 4. Overall survival (A) and recurrence-free survival (B) of patients undergoing surgical resection due to alveolar echinococcosis.

a follow-up of 76 months in the OH cohort (p = 0.245). Fig 4B shows that the 5-year recurrence-free survival was not different (LH 100% vs. OH 96%, p = 0.391). However, recurrence was seen after a mean time of 68 months after resection. All patients with recurrence (n = 3/70, 4%) in the OH cohort recurred after minor resections. Whereas one patient had R0 resection, the other two patients had missing data because they were treated in the earlier days. Postoper-ative treatment with albendazole was administered at least in 2 patients, whereas the other patient had missing data. Additionally, one of the patients developed increasing liver enzymes and the treatment needed to be stopped after 5 months. In all patients recurrent lesions were apparent at the former resection margin. Mean diameter of lesions in patients with recurrence was 5.0 cm, compared to 7.0 cm in patients without recurrence (p = 0.867) and distant extrahepatic lesion was present in one patient. However, detailed pathological data of recurrence in OH cohort was incomplete (n = 2) due to missing reports.

#### Subgroup analysis comparing PNM stage 1 patients

After subgroup analysis of all patients with PNM stage 1 showed that 18 patients (n = 18/43, 42%) were treated with laparoscopic approach, whereas 25 patients (n = 25/43, 58%) with limited disease stage had open hepatectomy (Table 3). The patients of the PNM stage 1 LH cohort had slightly more female patients (56% [n = 10/18] vs. 48% [n = 12/25], p = 0.629) and were older (61 years vs. 53 years, p = 0.076). Comparing mean maximal diameter of the AE lesion, the PNM stage 1 LH cohort had smaller lesions (3.3 cm vs. 6.0 cm, p = 0.008). Distribution pattern showed following segments affected by liver lesions in the LH cohort: One left-sided segment (n = 2/18, 11%), one central segment (n = 1/18, 6%), one right sided segment (n = 2/18, 11%), two left-sided segments (n = 2/18, 11%), two central segments (n = 1/18, 6%), two rightsided segments (n = 2/18, 11%), two bilateral segments (n = 1/18, 6%), more than three rightsided segments (n = 1/18, 6%) and more than three bilateral segments (n = 6/18, 33%). In contrast, in the OH cohort hepatic lesions were located in one central segment (n = 2/25, 8%), one right sided segment (n = 4/25, 16%), two left-sided segments (n = 1/25, 4%), two right-sided segments (n = 7/25, 28%), two bilateral segments (n = 2/25, 8%), more than three left-sided segments (n = 1/25, 4%), more than three right-sided segments (n = 4/25, 16%) and more than three bilateral segments (n = 4/25, 16%). Fig 5 shows that patients in the LH cohort were slightly more often treated for AE in left-lateral segments (p = 0.081), whereas patients in the OH cohort had more right liver resections (p = 0.163). The difference in operation time stayed significantly shorter with 120 minutes vs. 223 minutes (p < 0.001) in the subgroup analysis of PNM stage 1 patients. Postoperative major complication rate was low in both PNM stage 1

Variable	LH (n = 18)	OH (n = 25)	p-value
Gender, n (%)			0.629
Female	10/18 (56)	12/25 (48)	
Male	8/18 (44)	13/25 (52)	
Age, mean years (SD)	61 (±16)	53 (±16)	0.076
Maximal diameter of lesion, mean cm (SD)	3.3 (±1.9)	6.0 (±3.4)	0.008
Extrahepatic lesions, n (%)	0/18 (0)	0/25 (0)	-
Time from diagnosis to operation, mean months (SD)	4.4 (±5.5)	2.8 (±3.4)	0.324
Major operation, n (%)	0/18	13/25 (52)	0.003
Operation time, mean minutes (SD)	120 (±52)	223 (±52)	<0.001
Complications (Clavien-Dindo), n (%)	1/18 (6)	8/25 (28)	
1	0/18 (0)	0/25 (0)	-
2	0/18 (0)	3/25 (12)	0.132
3a	1/18 (6)	3/25 (12)	0.132
3b	0/18 (0)	2/25 (8)	0.759
4a	0/18 (0)	0/25 (0)	-
4b	0/18 (0)	0/25 (0)	-
_ 5	0/18 (0)	0/25 (0)	-
Major complication (Clavien-Dindo $\geq$ 3b), n (%)	0/18 (0)	2/25 (8)	0.759
Length of hospital stay, mean days (SD)	4 (±2)	12 (±7)	<0.001
Preoperative treatment with albendazol, n (%)	11/18 (61)	17/25 (68)	0.388
Postoperative treatment with albendazol, n (%)	12/18 (67)	22/25 (88)	0.163
Resection status, n (%)			
R0	12/18 (67)	19/25 (76)	0.506
R1	4/18 (22)	3/25 (12)	0.376
R2	0/18 (0)	0/25 (0)	-
Unknown	2/18 (11)	3/25 (12)	0.929
Recurrence, n (%)	0/18 (0)	2/25 (8)	0.682

Table 3. Comparison of demographic and perioperative clinical data between laparoscopic and open resection of PNM I staged patients with hepatic alveolar echinococcosis.

LH, Laparoscopic hepatectomy; OH, open hepatectomy; SD, standard deviation

https://doi.org/10.1371/journal.pntd.0010708.t003

cohorts (PNM stage 1 LH 0% [n = 0/18] vs. PNM stage 1 OH 8% [n = 2/25], p = 0.759). Mean length of stay was shorter in PNM stage 1 LH cohort (4 days vs. 12 days, p < 0.001). R0-resection was performed in 67% (n = 12/18) in the PNM stage 1 LH cohort and 76% (n = 19/25) in the PNM stage 1 OH cohort respectively (p = 0.569). Mean resection margin of R0 resected patients was insignificant smaller in the PNM stage 1 LH cohort (PNM stage 1 LH 0.3 cm [±0.4 cm, 10/12] vs. PNM stage 1 OH 0.5 cm [±0.9 cm, 18/19], p = 0.621). Recurrence rate during follow-up was low (PNM stage 1 LH 0% [n = 0/18] vs. PNM stage 1 OH 8% [n = 2/25], p = 0.682).

#### Discussion

In this single-center retrospective study of alveolar echinococcosis (AE) patients, laparoscopic hepatic resection of AE was found to be feasible and safe, with no signs of disease recurrence after mean follow-up of 55 months.

While both cohorts of laparoscopic and open resections had similar baseline characteristics in terms of gender, age, time to resection and preoperative benzimidazole (BZM) therapy with (ABZ), patients operated by laparoscopic resection had significantly lower P-stages, smaller



Fig 5. Location of hepatic AE lesions according to surgical approach of PNM stage 1 patients.

hepatic tumors and no extrahepatic disease. This points to a potential selection bias towards less advanced and easier to resect patients in our study. In addition to the learning curve that comes with the implementation of a new surgical technique, it needs to be taken into account that advanced AE often grows along the central bile ducts requiring complex biliary reconstructions more difficult to perform with the laparoscopic approach [27]. Nevertheless, our subgroup analysis of PNM stage 1, where baseline characteristics of the laparoscopic resected patients approach those of the open resected patients even more, supports our results.

Similar to the steady increase of laparoscopic liver resections for other indications [16,28,29], we observed a continuous rise of laparoscopic AE resections over time with a shorter hospital stay and significantly lower complication rates reflecting the well-known benefits of laparoscopic liver surgery. Only one larger series of 13 laparoscopically resected AE-patients was published reporting a major complication rate of 7.7% [13]. Thus, our data might be also compared with the outcome of other cohorts of oncologic laparoscopic liver surgery, with major complication rates of around 11% [30]. Perioperative data of our cohort showed a significantly shorter operation time which is comparable to studies of laparoscopic resection of colorectal liver metastasis [31,32], also taking into account a potential bias towards easier resections.

Short operating times and low complication rates clearly result in significantly shorter hospital stays, which has already been demonstrated for oncologic laparoscopic liver resections [28,30,33]. Complication rates and length of stay of our open hepatectomy cohort were comparable with those from other centers performing AE-resections [9,10,34].

In our laparoscopic hepatectomy cohort fewer patients received postoperative ABZ therapy and if administered the duration of therapy was shorter than the 24 months postoperative treatment recommended by international standards [5,35]. Despite these recommendations, we did not observe the development of metastasis or recurrences after complete laparoscopic resection, but the rather short follow-up should be taken into consideration when interpreting these results. In contrast, the open hepatectomy cohort shows 3 patients with recurrence. However, these patients had either extent disease or where treated in the early days, where experience in AE treatment was lower. Furthermore, the resection margin status of two patients was not available, which makes it impossible to draw any conclusion for these patients. With the simultaneous improvement of disease activity evaluation by our serology [26] and our collected monocentric experience with very low recurrence rates after AE resection, we have now developed a disease surveillance approach based on radiological examinations and activity measurement by recEm18 serology guiding the indication and duration of postoperative ABZ treatment. This approach, which is based on retrospective data, has currently not been validated yet and clearly requires additional prospective validation.

Perioperative serum antibody levels against recEm18, Em2 and EgHF antigens were routinely analyzed during diagnostic work-up and follow-up in our center during the last years [22,26]. It allows a differentiation between Echinococcus species during diagnostic work-up [36] and due to the negativization after curative resection it may act as an alternative in disease surveillance. However, the possible role of recEm18 in disease surveillance needs further prospective studies. Therefore, assessment of recEm18 serum antibody levels is of most clinical use of all mentioned tests so far [8,22]. In line with the study by Gottstein et al. [26], all 3 ELI-SAs showed decreased values after resection suggesting a correlation with the load of parasites. Em2 has more cross-reactions with cystic echinococcosis [37] and is used as a diagnostic tool just in combination with EgHF and recEm18. The fact, that the LH cohort had lower values of recEm18 compared to the OH cohort might be explained by the smaller and less active lesions.

The rates of histologically margin-positive resections (R1) were not different between laparoscopic and open resection in our cohort. While conventional standards recommend to perform resections with a safety margin of 2 cm [5,11], others have reported that even lower margins of up to 1mm showed good long term outcomes in combination with anthelminthic therapy [38]. In line of the study by Hillenbrand et al., even patients with microscopically positive resection margins in our LH cohort did not develop disease recurrence during follow-up. While the slightly higher recurrence rate of 4% in our OH cohort is clearly in the range of published recurrence rates between 2% and 16% at 5–20 years after curative resection [9,11,14,19], the resection margin seemed to have no influence on disease recurrence. 5-year recurrencefree survival and overall survival rates of our patients were similar to the current literature after curative resection [9–11,14] and no differences between open or laparoscopic surgery could be detected.

Even though our study has clear limitations due to its retrospective design, missing histopathological data in patients with recurrence, which avoids a differentiation between recurrence and persistence of disease, a certain selection bias towards less advanced patients for the laparoscopic approach and a limited follow-up, we are the first western group presenting results of the largest cohort of laparoscopic AE resections so far. Laparoscopic early-stage AE resection is feasible and save with long-term recurrence-free survival.

# Conclusion

Laparoscopic resection of AE seems to be feasible and safe with no negative impact on perioperative outcomes, early disease recurrence or survival within a follow-up of mean 55 months. Therefore each patient should be evaluated by an experienced minimal invasive liver surgery team at an AE center with interdisciplinary treatment, eventually also allowing major laparoscopic AE resections in the future.

# Acknowledgments

The authors wish to thank Prof. Dr. Bruno Gottstein for his expertise in serology and critical review of the manuscript. Further, we would like to thank Dr. Magdalena Skowronska for providing us detailed pathological information.

# **Author Contributions**

Conceptualization: Severin Gloor, Guido Beldi, Anja Lachenmayer.

Data curation: Severin Gloor, Anja Lachenmayer.

Formal analysis: Severin Gloor.

Investigation: Severin Gloor.

Methodology: Severin Gloor, Guido Beldi, Anja Lachenmayer.

Project administration: Anja Lachenmayer.

Resources: Severin Gloor, Guido Beldi, Anja Lachenmayer.

Software: Severin Gloor, Anja Lachenmayer.

Supervision: Daniel Candinas.

Validation: Guido Beldi, Anja Lachenmayer.

Visualization: Severin Gloor, Anja Lachenmayer.

Writing – original draft: Severin Gloor, Anja Lachenmayer.

Writing - review & editing: Daniel Candinas, Guido Beldi.

#### References

- Ammann RW, Eckert J. Cestodes. Echinococcus. Gastroenterology clinics of North America. 1996; 25 (3):655–89. https://doi.org/10.1016/s0889-8553(05)70268-5 PMID: 8863045
- Kern P, Bardonnet K, Renner E, Auer H, Pawlowski Z, Ammann RW, et al. European echinococcosis registry: human alveolar echinococcosis, Europe, 1982–2000. Emerging infectious diseases. 2003; 9 (3):343–9. Epub 2003/03/20. https://doi.org/10.3201/eid0903.020341 PMID: 12643830; PubMed Central PMCID: PMC2958541.
- Torgerson PR, Schweiger A, Deplazes P, Pohar M, Reichen J, Ammann RW, et al. Alveolar echinococcosis: from a deadly disease to a well-controlled infection. Relative survival and economic analysis in Switzerland over the last 35 years. Journal of hepatology. 2008; 49(1):72–7. <u>https://doi.org/10.1016/j.jhep.2008.03.023</u> PMID: 18485517.
- Gottstein B, Stojkovic M, Vuitton DA, Millon L, Marcinkute A, Deplazes P. Threat of alveolar echinococcosis to public health—a challenge for Europe. Trends in parasitology. 2015; 31(9):407–12. <a href="https://doi.org/10.1016/j.pt.2015.06.001">https://doi.org/10.1016/j.pt.2015.06.001</a> PMID: 26115902.
- Brunetti E, Kern P, Vuitton DA, Writing Panel for the W-I. Expert consensus for the diagnosis and treatment of cystic and alveolar echinococcosis in humans. Acta tropica. 2010; 114(1):1–16. <u>https://doi.org/ 10.1016/j.actatropica.2009.11.001</u> PMID: 19931502.
- Graeter T, Ehing F, Oeztuerk S, Mason RA, Haenle MM, Kratzer W, et al. Hepatobiliary complications of alveolar echinococcosis: A long-term follow-up study. World journal of gastroenterology. 2015; 21 (16):4925–32. https://doi.org/10.3748/wjg.v21.i16.4925 PMID: 25945006; PubMed Central PMCID: PMC4408465.
- Muller N, Frei E, Nunez S, Gottstein B. Improved serodiagnosis of alveolar echinococcosis of humans using an in vitro-produced Echinococcus multilocularis antigen. Parasitology. 2007; 134(Pt 6):879–88. https://doi.org/10.1017/S0031182006002083 PMID: 17306060.

- Tappe D, Frosch M, Sako Y, Itoh S, Gruner B, Reuter S, et al. Close relationship between clinical regression and specific serology in the follow-up of patients with alveolar echinococcosis in different clinical stages. The American journal of tropical medicine and hygiene. 2009; 80(5):792–7. PMID: 19407126.
- Joliat GR, Melloul E, Petermann D, Demartines N, Gillet M, Uldry E, et al. Outcomes After Liver Resection for Hepatic Alveolar Echinococcosis: A Single-Center Cohort Study. World journal of surgery. 2015; 39(10):2529–34. https://doi.org/10.1007/s00268-015-3109-2 PMID: 26067633.
- Chen KF, Tang YY, Wang R, Fang D, Chen JH, Zeng Y, et al. The choose of different surgical therapies of hepatic alveolar echinococcosis: A single-center retrospective case-control study. Medicine. 2018; 97(8):e0033. https://doi.org/10.1097/MD.00000000010033 PMID: 29465544; PubMed Central PMCID: PMC5841995.
- Buttenschoen K, Carli Buttenschoen D, Gruener B, Kern P, Beger HG, Henne-Bruns D, et al. Longterm experience on surgical treatment of alveolar echinococcosis. Langenbeck's archives of surgery. 2009; 394(4):689–98. https://doi.org/10.1007/s00423-008-0392-5 PMID: 18651165.
- Filippou D, Tselepis D, Filippou G, Papadopoulos V. Advances in liver echinococcosis: diagnosis and treatment. Clinical gastroenterology and hepatology: the official clinical practice journal of the American Gastroenterological Association. 2007; 5(2):152–9. https://doi.org/10.1016/j.cgh.2006.08.017 PMID: 17157079.
- Wan L, Ran B, Aji T, Shalayiadang P, Jiang T, Shao Y, et al. Laparoscopic hepatectomy for the treatment of hepatic alveolar echinococcosis. Parasite. 2021; 28:5. https://doi.org/10.1051/parasite/2021001 PMID: 33439119; PubMed Central PMCID: PMC7805388.
- Kawamura N, Kamiyama T, Sato N, Nakanishi K, Yokoo H, Kamachi H, et al. Long-term results of hepatectomy for patients with alveolar echinococcosis: a single-center experience. Journal of the American College of Surgeons. 2011; 212(5):804–12. https://doi.org/10.1016/j.jamcollsurg.2011.02.007 PMID: 21398158.
- Zhao ZM, Yin ZZ, Meng Y, Jiang N, Ma ZG, Pan LC, et al. Successful robotic radical resection of hepatic echinococcosis located in posterosuperior liver segments. World journal of gastroenterology. 2020; 26 (21):2831–8. https://doi.org/10.3748/wjg.v26.i21.2831 PMID: 32550758; PubMed Central PMCID: PMC7284188.
- Buell JF, Cherqui D, Geller DA, O'Rourke N, Iannitti D, Dagher I, et al. The international position on laparoscopic liver surgery: The Louisville Statement, 2008. Annals of surgery. 2009; 250(5):825–30. https://doi.org/10.1097/sla.0b013e3181b3b2d8 PMID: 19916210.
- Sato N, Uchino J, Takahashi M, Aoki S, Takahashi H, Yamashita K, et al. Surgery and outcome of alveolar echinococcosis of the liver: historical comparison of mass screening systems in Japan. International surgery. 1997; 82(2):201–4. Epub 1997/04/01. PMID: 9331854.
- Gruner B, Kern P, Mayer B, Grater T, Hillenbrand A, Barth TEF, et al. Comprehensive diagnosis and treatment of alveolar echinococcosis: A single-center, long-term observational study of 312 patients in Germany. GMS Infect Dis. 2017; 5:Doc01. Epub 2017/01/06. https://doi.org/10.3205/id000027 PMID: 30671323; PubMed Central PMCID: PMC6301735.
- Beldi G, Vuitton D, Lachenmayer A, Heyd B, Dufour JF, Richou C, et al. Is ex vivo liver resection and autotransplantation a valid alternative treatment for end-stage hepatic alveolar echinococcosis in Europe? Journal of hepatology. 2019; 70(5):1030–1. <u>https://doi.org/10.1016/j.jhep.2018.12.011</u> PMID: 30718093.
- Venkatesan P. Albendazole. J Antimicrob Chemother. 1998; 41(2):145–7. Epub 1998/04/09. <a href="https://doi.org/10.1093/jac/41.2.145">https://doi.org/10.1093/jac/41.2.145</a> PMID: 9533454.
- Horton J. Albendazole for the treatment of echinococcosis. Fundamental & clinical pharmacology. 2003; 17(2):205–12. https://doi.org/10.1046/j.1472-8206.2003.00171.x PMID: 12667231.
- Lachenmayer A, Gebbers D, Gottstein B, Candinas D, Beldi G. Elevated incidence of alveolar echinococcosis in immunocompromised patients. Food and Waterborne Parasitology. 2019;16. <u>https://doi.org/10.1016/j.fawpar.2019.e00060 PMID: 32095630</u>
- Kern P, Wen H, Sato N, Vuitton DA, Gruener B, Shao Y, et al. WHO classification of alveolar echinococcosis: principles and application. Parasitology international. 2006; 55 Suppl:S283–7. Epub 2005/12/14. https://doi.org/10.1016/j.parint.2005.11.041 PMID: 16343985.
- Ban D, Tanabe M, Ito H, Otsuka Y, Nitta H, Abe Y, et al. A novel difficulty scoring system for laparoscopic liver resection. Journal of hepato-biliary-pancreatic sciences. 2014; 21(10):745–53. Epub 2014/ 09/23. https://doi.org/10.1002/jhbp.166 PMID: 25242563.
- Clavien PA, Barkun J, de Oliveira ML, Vauthey JN, Dindo D, Schulick RD, et al. The Clavien-Dindo classification of surgical complications: five-year experience. Annals of surgery. 2009; 250(2):187–96. https://doi.org/10.1097/SLA.0b013e3181b13ca2 PMID: 19638912.

- Gottstein B, Lachenmayer A, Beldi G, Wang J, Merkle B, Vu XL, et al. Diagnostic and follow-up performance of serological tests for different forms/courses of alveolar echinococcosis. Food Waterborne Parasitol. 2019; 16:e00055. https://doi.org/10.1016/j.fawpar.2019.e00055 PMID: 32095626; PubMed Central PMCID: PMC7034017.
- Wen H, Vuitton L, Tuxun T, Li J, Vuitton DA, Zhang W, et al. Echinococcosis: Advances in the 21st Century. Clinical microbiology reviews. 2019;32(2). https://doi.org/10.1128/CMR.00075-18 PMID: 30760475; PubMed Central PMCID: PMC6431127.
- Tian ZQ, Su XF, Lin ZY, Wu MC, Wei LX, He J. Meta-analysis of laparoscopic versus open liver resection for colorectal liver metastases. Oncotarget. 2016; 7(51):84544–55. <u>https://doi.org/10.18632/oncotarget.13026</u> PMID: <u>27811369</u>; PubMed Central PMCID: PMC5356680.
- Levi Sandri GB, Spoletini G, Vennarecci G, Francone E, Abu Hilal M, Ettorre GM. Laparoscopic liver resection for large HCC: short- and long-term outcomes in relation to tumor size. Surgical endoscopy. 2018; 32(12):4772–9. https://doi.org/10.1007/s00464-018-6225-x PMID: 29770883.
- Fretland AA, Kazaryan AM, Bjornbeth BA, Flatmark K, Andersen MH, Tonnessen TI, et al. Open versus laparoscopic liver resection for colorectal liver metastases (the Oslo-CoMet Study): study protocol for a randomized controlled trial. Trials. 2015; 16:73. https://doi.org/10.1186/s13063-015-0577-5 PMID: 25872027; PubMed Central PMCID: PMC4358911.
- Belli G, Limongelli P, Fantini C, D'Agostino A, Cioffi L, Belli A, et al. Laparoscopic and open treatment of hepatocellular carcinoma in patients with cirrhosis. The British journal of surgery. 2009; 96(9):1041–8. https://doi.org/10.1002/bjs.6680 PMID: 19672933.
- 32. Koffron AJ, Auffenberg G, Kung R, Abecassis M. Evaluation of 300 minimally invasive liver resections at a single institution: less is more. Annals of surgery. 2007; 246(3):385–92; discussion 92–4. https:// doi.org/10.1097/SLA.0b013e318146996c PMID: 17717442; PubMed Central PMCID: PMC1959347.
- 33. Montalti R, Berardi G, Laurent S, Sebastiani S, Ferdinande L, Libbrecht LJ, et al. Laparoscopic liver resection compared to open approach in patients with colorectal liver metastases improves further resectability: Oncological outcomes of a case-control matched-pairs analysis. European journal of surgical oncology: the journal of the European Society of Surgical Oncology and the British Association of Surgical Oncology. 2014; 40(5):536–44. https://doi.org/10.1016/j.ejso.2014.01.005 PMID: 24555996.
- Du C, Liu Z, Yang X, Yan L, Li B, Wen T, et al. Hepatectomy for patients with alveolar echinococcosis: Long-term follow-up observations of 144 cases. International journal of surgery. 2016; 35:147–52. Epub 2016/10/25. https://doi.org/10.1016/j.ijsu.2016.09.094 PMID: 27693514.
- Guidelines for treatment of cystic and alveolar echinococcosis in humans. WHO Informal Working Group on Echinococcosis. Bulletin of the World Health Organization. 1996; 74(3):231–42. PMID: 8789923; PubMed Central PMCID: PMC2486920.
- Gottstein B, Eckert J, Fey H. Serological differentiation between Echinococcus granulosus and E. multilocularis infections in man. Z Parasitenkd. 1983; 69(3):347–56. Epub 1983/01/01. https://doi.org/10. 1007/BF00927876 PMID: 6192602.
- 37. Kronenberg PAD A.; Gottstein B.; Grimm F.; Müllhaupt B.; Meyer zu Schwabedissen C.; Aitbaev S.; Omorov R.A.; Abdykerimov K.K.; Minbaeva G.; Usubalieva J.; Siles-Lucas M.; Pepe P.; Rinaldi L.; Spiliotis M.; Wang J.; Müller N.; Torgerson P.R.; Deplazes P. Serological Assays for Alveolar and Cystic Echinococcosis—A Comparative Multi-Test Study in Switzerland and Kyrgyzstan. Pathogens. 2022; 11 (518).
- Hillenbrand A, Gruener B, Kratzer W, Kern P, Graeter T, Barth TF, et al. Impact of Safe Distance on Long-Term Outcome After Surgical Therapy of Alveolar Echinococcosis. World journal of surgery. 2017; 41(4):1012–8. https://doi.org/10.1007/s00268-016-3813-6 PMID: 27822723.